Author's response to reviews

Title: Administration of fibrinogen concentrate for refractory bleeding in massively transfused, non-trauma patients with coagulopathy: A retrospective study with comparator group

Authors:

Santiago R Leal-Noval (srlealnoval@gmail.com)
Manuel Casado (casadomanu@gmail.com)
Victoria Arellano-Orden (victoria_are_or@hotmail.com)
Reginald Dusseck (rdusseck@gmail.com)
Javier Bautista (francisco.j.bautista.sspa@juntadeandalucia.es)
Manuel Múñoz-Gómez (mmunoz@uma.es)
José Naranjo-Izurieta (jose.naranjo.10@gmail.com)
Antonio M Puppo-Moreno (apuppo@telefonica.net)
Aurelio Cayuela (aurelio.cayuela.sspa@juntadeandalucia.es)

Version: 7 Date: 7 October 2014

Author's response to reviews: see over
Seville, October 7, 2014.

Dear Sir,

Please, find enclosed our r-revised version of the manuscript entitled “Administration of fibrinogen concentrate for refractory bleeding in massively transfused, non-trauma patients with coagulopathy” which we would like to be considered for publication in BMC Anesthesiology.

According with your recommendations, we have performed the following changes:

1. Point by point response to the most recent comments of the referees. Please, see below

Please, let me know if any additional information is needed

We appreciate the opportunity for our manuscript to be peer-reviewed and look forward to your decision at your earlier convenience.

Sincerely,

S. R. Leal-Noval, MD, PhD

Point by point response to the most recent comments of the referees

Reviewer's report
Title: Administration of fibrinogen concentrate for refractory bleeding in massively transfused, non-trauma patients with coagulopathy A retrospective study with comparator group

**Reviewer number: 3**
Referee's comments to the author(s)

Minor essential revision
One issue that was addressed in my first review - namely the discrepancy between preFIB-values in the text in the result section and in table 2 - has not been commented. The text says:
"After administering FBNc, fibrinogen levels increased from 1.3 [0.9, 2.3] g/L to
In table two it says that Pre-FBNc is 1.6 and Post-FBNc is 1.8. Is there something I have misunderstood? Consider correcting or explaining.

**Answer**

We have corrected this mistake, and added the real values on table 2 (please, see page 22).